\documentclass[11pt,twoside]{article}
\usepackage{amsmath, amsthm, amscd, amsfonts, amssymb, graphicx, color}
\usepackage[bookmarksnumbered, colorlinks]{hyperref} \usepackage{float}
%\usepackage{lipsum}
\usepackage{afterpage}
\usepackage[labelfont=bf]{caption}
%\usepackage[nottoc,notlof,notlot]{tocbibind}
%\renewcommand\bibname{References}
\def\bibname{\Large \bf  References}
%\usepackage{lipsum}
\usepackage{fancyhdr}
\pagestyle{fancy}
\fancyhf{}
\renewcommand{\headrulewidth}{0pt}
\fancyhead[LE,RO]{\thepage}
\thispagestyle{empty}
%\afterpage{\lhead{new value}}

\fancyhead[CE]{K. AKHAVAN ZAKERI, M.A. FARIBORZI ARAGHI AND S. ZIARI}
\fancyhead[CO]{ITERATIVE APPROACH FOR A CLASS OF FUZZY VOLTERRA ...}



%\topmargin=-1.6cm
\textheight 17.5cm%
\textwidth  12cm %
\topmargin   8mm  %
\oddsidemargin   20mm   %
\evensidemargin   20mm   %
\footskip=24pt     %

\newtheorem{theorem}{Theorem}[section]
\newtheorem{lemma}[theorem]{Lemma}
\newtheorem{proposition}[theorem]{Proposition}
\newtheorem{corollary}[theorem]{Corollary}
\theoremstyle{definition}
\newtheorem{definition}[theorem]{Definition}
\newtheorem{example}[theorem]{Example}
\newtheorem{xca}[theorem]{Exercise}
%\theoremstyle{remark}
\newtheorem{remark}[theorem]{Remark}
\renewenvironment{proof}{{\bfseries \noindent Proof.}}{~~~~$\square$}
\makeatletter
\def\th@newremark{\th@remark\thm@headfont{\bfseries}}
\makeatletter





%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
% If you want to insert other packages. Insert them here
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

%\long\def\symbolfootnote[#1]#2{\begingroup%
%\def\thefootnote{\fnsymbol{footnote}}\footnote[#1]{#2}\endgroup}



 \def \thesection{\arabic{section}}


\begin{document}
\begin{thebibliography}{9}

\bibitem {Abbas2} S. Abbasbandy, E. Babolian and M. Alavi, Numerical method for solving linear Fredholm
fuzzy integral equations of the second kind. {\it Chaos Solitons \& Fractals}, 31 (2007), 138-146.

\bibitem {Akhavan} K. Akhavan Zakeri, S. Ziari, M.A. Fariborzi Araghi and I. Perfilieva,
Efficient Numerical Solution to a Bivariate Nonlinear Fuzzy Fredholm Integral Equation.
{\it IEEE Transactions on Fuzzy Systems}, 29(2) (2021), 442-454.

\bibitem {Babol} E. Babolian, H. Sadeghi Goghary and S. Abbasbandy, Numerical solution of linear Fred-
holm fuzzy integral equations of the second kind by Adomian method. {\it  Appl. Math.
Comput.},  161 (2005), 733-744.

\bibitem {Agrawal} R.P. Agarwal, D. Oregan and B.V. Lakshmikantham, Fuzzy
Volterra integral equations: A stacking theorem approach,
{\it Applicable Analysis}, 83(5) (2004), 521-532.

\bibitem {Balach1} K. Balachandran, K. Kanagarajan, Existence of solutions of
general nonlinear fuzzy Volterra-Fredholm integral equations, {\it J. Appl. Math.
Stochastic Anal.} 3 (2005), 333-343.

\bibitem {Balach2} K. Balachandran and P. Prakash,  Existence of solutions of
nonlinear fuzzy Volterra-Fredholm integral equations, {\it Indian J. Pure Appl.
Math.}, 33 (2002), 329-343.

\bibitem {Bede} B. Bede and S.G. Gal,  Quadrature rules for integrals of
fuzzy-valued functions, {\it Fuzzy Sets and Systems}, 145 (2004), 359-380.

\bibitem {Bica1} A.M. Bica,  Error estimation in the approximation of the
solution of nonlinear fuzzy Fredholm integral equations, {\it Information Sciences},
178 (2008), 1279-1292.

\bibitem {Bica2} A.M. Bica and C. Popescu,  Approximating the solution of nonlinear
Hammerstein fuzzy integral equations, {\it Fuzzy Sets and Systems}, 245 (2014), 1-17.

\bibitem {Bica3} A.M. Bica and C. Popescu, Iterative numerical method for nonlinear
fuzzy Volterra integral equations, {\it Journal of Intelligent \& Fuzzy Systems},
32(3) (2017), 1639-1648.

\bibitem {cong2} Congxin Wu and Gong Zengtai,  On Henstock integral of
fuzzy-valued functions (I), {\it Fuzzy Sets Syst.}, 120 (2001), 523-532.

\bibitem {Diamond} P. Diamond,  Theory and applications of fuzzy Volterra integral
equations, {\it IEEE Transactions on Fuzzy Systems}, 10(1) (2002), 97-102.

\bibitem {Ezati1} R. Ezzati and S. Ziari, Numerical solution and error estimation of
fuzzy Fredholm integral equation using fuzzy Bernstein polynomials,
{\it Aust J Basic Appl Sci}, 5(9) (2011), 2072-2082.

\bibitem {Ezati2} R. Ezzati and S. Ziari,  Numerical solution of nonlinear fuzzy
Fredholm integral equations using iterative method, {\it Appl. Math. Comput.}, 225
(2013), 33-42.

\bibitem {Fariborzi} M.A. Fariborzi Araghi and N. Parandin, Numerical solution of fuzzy Fredholm integral
equations by the Lagrange interpolation based on the extension principle. {\it Soft Comput.},
15 (2011), 2449-2456.

\bibitem {Friedman3} M. Friedman, M. Ma and A. Kandel,  Solutions to fuzzy integral
equations with arbitrary kernels, {\it International Journal of Approximate
Reasoning}, 20 (1999), 249-262.

\bibitem {Friedman4} M. Friedman, M. Ma and A. Kandel,  Numerical solutions of
fuzzy differential and integral equations, {\it Fuzzy Sets and Systems}, 106 (1999), 35-48.

\bibitem {Gal} S.G. Gal,  {\it Approximation theory in fuzzy setting, in: G.A.
Anastassiou (Ed.), Handbook of Analytic-Computational Methods in Applied
Mathematics}, Chapman \& Hall, CRC Press, Boca Raton, London, New York,
Washington DC, 2000 (Chapter 13).

\bibitem {Goetschel} R. Goetschel and W. Voxman,  Elementary fuzzy calculus, {\it Fuzzy
Sets and Systems}, 18 (1986), 31-43.

\bibitem{Jiang} ZH. Jiang and W. Schanfelberger, {\it Block-Pulse Functions and Their
Applications in Control Systems}, Springer, Berlin (1992).

\bibitem {Kaleva} O. Kaleva,  Fuzzy differential equations, {\it Fuzzy Sets and
Systems}, 24 (1987), 301-317.

\bibitem {Mordeson} J. Mordeson and W. Newman,  Fuzzy integral equations, {\it Inform.
Sci.}, 87 (1995), 215-229.

\bibitem {Nanda} S. Nanda,  On integration of fuzzy mappings, {\it Fuzzy Sets and
Systems}, 32 (1989), 95-101.

\bibitem {Nieto} J.J. Nieto and R. Rodriguez-Lopez, Bounded solutions for fuzzy differential and integral
equations, {\it Chaos Soliton Fract.}, 27(5) (2006), 1376-1386.

\bibitem {Parandin} N. Parandin and M.A. Fariborzi Araghi, The numerical solution of linear fuzzy Fredholm
integral equations of the second kind by using finite and divided differences methods,
{\it Soft Comput.}, 15 (2010), 729-741.

\bibitem {Park1} J.Y. Park, S.Y. Lee and J.U. Jeong, The approximate solution of
fuzzy functional integral equations, {\it Fuzzy Sets and Systems}, 110 (2000), 79-90.

\bibitem {Park2} J.Y. Park and J.U. Jeong,  On the existence and uniqueness of
solutions of fuzzy Volttera-Fredholm integral equations, {\it Fuzzy Sets and
Systems}, 115 (2000), 425-431.

\bibitem {Park3} J.Y. Park and H.K. Han,  Existence and uniqueness theorem for a
solution of fuzzy Volterra integral equations, {\it Fuzzy Sets and Systems},
105 (1999), 481-488.

\bibitem {Park4} J.Y. Park and J. U. Jong,  A note on fuzzy integral equations, {\it Fuzzy Sets and Systems},
108 (1999), 193-200.

\bibitem {Song} S. Song, Q.-Y. Liu and Q.-C. Xu,  Existence and comparison
theorems to Volterra fuzzy integral equation in $(E^n,D)$, {\it Fuzzy Sets and
Systems}, 104 (1999), 315-321.

\bibitem {Seikala} S. Seikkala,  On the fuzzy initial value problem, {\it Fuzzy Sets Syst.}, 24 (1987), 319-330.

\bibitem {Subra} P.V. Subrahmanyam and S.K. Sudarsanam,  A note on fuzzy
Volterra integral equations, {\it Fuzzy Sets Syst}, 81 (1996), 237-240.

\bibitem {C.Wu} C.Wu and Z. Gong,  On Henstock integral of fuzzy-number-valued
functions (I), {\it Fuzzy Sets and Systems}, 120 (2001), 523-532.

\bibitem {HC.Wu} H.-C. Wu,  The fuzzy Riemann integral and its numerical
integration, {\it Fuzzy Sets and Systems}, 110 (2000), 1-25.

\bibitem {Ziari1} S. Ziari, R. Ezzati and S. Abbasbandy, Numerical solution of linear fuzzy Fredholm inte-
gral equations of the second kind using fuzzy Haar wavelet, {\it In: Advances in Computa-
tional Intelligence. Communications in Computer and Information Science}, 299 (2012), 79-89.

\bibitem {Ziari2} S. Ziari and A.M. Bica,  New error estimate in the iterative
numerical method for nonlinear fuzzy Hammerstein-Fredholm integral equations,
{\it Fuzzy Sets and Systems}, 295 (2016), 136-152.

\bibitem {Ziari3} S. Ziari and R. Ezzati, Fuzzy Block-Pulse Functions and Its Application to Solve Linear Fuzzy
Fredholm Integral Equations of the Second Kind, {\it In: Carvalho J., Lesot MJ., Kaymak U., Vieira S.,
Bouchon-Meunier B., Yager R. (eds) Information Processing and Management of Uncertainty in Knowledge-Based Systems,
IPMU 2016, Communications in Computer and Information Science}, vol 611 (2016), 821-832, Springer, Cham.

\bibitem {Ziari4} S. Ziari, Towards the accuracy of iterative numerical methods for
fuzzy Hammerstein-Fredholm integral equations, {\it Fuzzy Sets Syst.}, 375 (2019), 161-178.

\bibitem {Ziari5} S. Ziari and A.M. Bica, An iterative numerical method to solve nonlinear fuzzy Volterra-Hammerstein integral equations,
 {\it Journal of Intelligent \& Fuzzy Systems}, 37(5) (2019), 6717-6729.

\bibitem {Ziari6} S. Ziari, I. Perfilieva and S. Abbasbandy, Block-pulse functions in the method of successive approximations
for nonlinear fuzzy Fredholm integral equations, {\it Differ Equ Dyn Syst.}, (2019) https://doi.org/10.1007/s12591-019-00482-y.

\end{thebibliography}

%\baselineskip 9mm
%\setcounter{page}{}
\thispagestyle{plain}
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
{\noindent Journal of ... \\  %Mathematical Extension \\
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
Vol. XX, No. XX, (2020), pp-pp (Will be inserted by layout editor)}\\
ISSN: 1735-8299\\
URL: http://www.ijmex.com\\
‎\vspace*{9mm}
‎
\begin{center}

{\Large \bf
Iterative Approach for a Class of Fuzzy Volterra Integral Equations Using Block Pulse Functions\\}
%{\bf Do You Have a Subtitle? \\ If so, Write It Here}


\let\thefootnote\relax\footnote{\scriptsize Received: Nov 2020; Accepted: Jun 2021 }

{\bf K. Akhavan Zakeri}\vspace*{-2mm}\\
\vspace{2mm} {\small  Central Tehran Branch,
Islamic Azad University} \vspace{2mm}

{\bf M.A. Fariborzi Araghi$^\dag$\let\thefootnote\relax\footnote{$^\dag$Corresponding Author}}\vspace*{-2mm}\\
\vspace{2mm} {\small  Central Tehran Branch,
Islamic Azad University} \vspace{2mm}

{\bf  S. Ziari$^*$\let\thefootnote\relax\footnote{$^*$Corresponding Author}}\vspace*{-2mm}\\
\vspace{2mm} {\small   Firoozkooh
Branch, Islamic Azad University} \vspace{2mm}

\end{center}

\vspace{4mm}


{\footnotesize
\begin{quotation}
{\noindent \bf Abstract.}
Fuzzy Integral equations is a mathematical tool for modeling the uncertain control system and economic.
In this paper, we present numerical solution of nonlinear fuzzy Volterra integral equations(NFVIEs) using successive approximations scheme and block-pulse functions. Additionally, the convergence analysis of the presented approach is investigated involving Lipschitz and several conditions and
error bound between the approximate and the exact solution is provided. Finally, to approve the outcomes concerned with the theory a numerical experiment is considered.
\end{quotation}
\begin{quotation}
\noindent{\bf AMS Subject Classification:} 03E72; 46S40; 47S40.

\noindent{\bf Keywords and Phrases:} Fuzzy Volterra integral equations,
Successive approximations scheme, $L$-Lipschitz fuzzy-valued function.
\end{quotation}}

\section{Introduction}
\label{intro} % It is advised to give each section and subsection a unique label.
Fuzzy integral equations is a significant tool for modeling and analysis of uncertain real-world problems
in various fields of science and engineering. The topic of fuzzy integral equations was first studied by
Kaleva (see \cite{Kaleva}), Seikkala (see \cite{Seikala}),
they converted the initial value problem for first order fuzzy differential equations to the fuzzy
Volterra integral equation.
In this regards, the distinct work on the fuzzy integral equations
is carried out by Mordeson and Newman (see \cite{Mordeson}) based on Zadeh's extension principle.
Afterwards, Diamond in \cite{Diamond} studied the NFVIE with applications in optimal control theory.
The existence and uniqueness results of the solution of fuzzy Volterra integral equations have been investigated by many researchers
using Banach fixed-point theorem (see \cite{Balach1,Balach2,Friedman3,Park2,Park4}).
%
In \cite{Nieto}, are provided conditions for having the bounded solutions of fuzzy integral equations.
%%
In the literature, we can find different approaches for solving fuzzy integral equation such as Picard's approximations scheme
and quadrature rules which are exploited in \cite{Akhavan,Bede,Bica1,Bica2,Bica3,Ezati2,Friedman3,Friedman4,Ziari2,Ziari3} and \cite{Park1}.
%%
In several papers are used direct method for solving fuzzy integral equations such as
the Adomian decomposition technique (see \cite{Babol}), Nystrom method (\cite{Abbas2}), Bernstein polynomials
and Haar wavelets (see \cite{Ezati1,Ziari1}).
%%
In some papers to solution of these equations are applied successive approximations scheme with
interpolation of Lagrange(see \cite{Fariborzi}), finite differences(see \cite{Parandin}) and block-pulse functions
(see \cite{Ziari3,Ziari6}).
%%
Bica and Popescu, in \cite{Bica3} presented an iterative scheme by using
successive approximation scheme and Trapezoidal quadrature formula for solving NFVIEs.
An iterative numerical procedure has been presented recently in \cite{Ziari5} which is solved NFVIE via three-point quadrature rule.\\
%
In the present study, we provide an iterative procedure to the NFVIE of the following form:%
\begin{equation}\label{eq1}
y(t)=g(t)\oplus (FR)\int_{a}^{t}f(t,s,y(s))ds,\qquad t\in [a,b]
\end{equation}%
where
$f:[a,b]\times [a,b]\times\mathbb{R}_{F}\rightarrow \mathbb{R}_{F}$ and
 $g:[a,b]\rightarrow \mathbb{R}_{F}$ are fuzzy continuous.
The purpose of the present work is to present an iterative numerical scheme to gain the approximate
solution of \eqref{eq1} based on block pulse functions.
%%
\section{Preliminaries}

Herein, the essential notions of fuzzy mathematics employed in the subsequent sections will be stated.
\begin{definition} \label{def2} \normalfont{(see \cite{C.Wu}.)}
\label{def1} A fuzzy number is a function $u$ from $\mathbb{R}$ to $\lbrack0,1]$ satisfying the following conditions:
\newline$(i)$ $u$ is normal,
\newline$(ii)$ $u$ is fuzzy convex set,%
\newline$(iii)$ $u$ is upper semi-continuous on $\mathbb{R}$,
\newline$(iv)$
$\overline{supp(u)}$ is a compact subset of $\mathbb{R}$, where $supp(u)=\{\,x\in \mathbb{R}:u(x)>0\}$ is the
support of $u$ and $\overline{supp(u)}$ denotes the closure of $supp(u)$.
\end{definition}
We denote by $\mathbb{R}_{F}$ the set of all fuzzy numbers.
%
For $0<r�\leq 1$, the $r-$level sets of $u$ is defined as $[u]^{r}=\{x\in \mathbb{R}:u(x)\geq r\}$.%
Then, we represent the parametric form of $u$ as:
\[
\lbrack u]^{r}=[\underline{u}(r),\overline{u}(r)],\quad 0<r�\leq 1,
\]
where $\underline{u}(r),\overline{u}(r)$ are two real functions $\underline{u},\overline{u}%
:[0,1]\rightarrow\mathbb{R},$ that $\underline{u}$ is increasing and $\overline{u}$ is
decreasing (see \cite{Goetschel}). Furthermore, $\lbrack u]^{0}=\overline{~\left\{x\in \mathbb{R}:u(x)>0\right\}}.$
So, the $r-$cuts of a fuzzy number $[u]^{r},$ $r\in\lbrack0,1],$ are compact subset of $\mathbb{R}$.
%
For $u_1,u_2\in \mathbb{R}_{F},$ $k\in\mathbb{R},$ the addition and the scalar
multiplication based on levelsetwise are defined as follows %
%%
\begin{flalign*}
& [u_1+u_2]^{r}=[\underline{u}_1(r)+\underline{u}_2(r),\overline{u}_1(r)+\overline{u}_2(r)%
], \forall 0\leq r�\leq 1\\
&\lbrack k\odot u_1]^{r}=\left\{
\begin{array}
[c]{c}%
\lbrack k\underline{u}_1(r),k\overline{u}_1(r)],\text{ \ if \ }k\geq0\\
\lbrack k\overline{u}_1(r),k\underline{u}_1(r)],\text{ \ if \ }k<0.
\end{array}
\right.
\end{flalign*}
%%
\begin{definition}
\label{def2} \normalfont{(see \cite{Gal}.)} Suppose that $u_1,u_2\in \mathbb{R}_{F}$ the value
$D(u_1,u_2)=\mathop{\sup }\limits_{r\in [0,1]} \max \{ |\underline{u}_1(r) -\underline{u}_2(r)| , |\overline{u}_1(r) -\overline{u}_2(r)|\, \} $
is the distance between $u_1$ and $u_2$.
\end{definition}
%
One can notice that $(\mathbb{R}_{F},D)$ is a complete metric space.
\begin{theorem}
\label{th3}(\cite{Gal}.) For any $u_1,~u_2,~u_3,~u\in \mathbb{R}_{F},~k_1\ge 0,~k_2\ge 0,~k\in \mathbb{R}$,
the following properties hold:
\newline$(i)$   $D(u_1\oplus u_3,u_2\oplus u_3)=D(u_1,u_2)$,
\newline$(ii)$  $D(k\odot u_1,k\odot u_2)=\left\vert k\right\vert D(u_1,u_2)$,
\newline$(iii)$ $D(u_1,u_2)\leq D(u_1,u_3)+D(u_3,u_2)$,
\newline$(iv)$  $D(u_1\oplus u_2,u_3)\leq D(u_1,u_3)+D(u_2,u_3)$,
\newline$(v)$   $D\left( k_{1}\cdot u,k_{2}\cdot u\right) =\left\vert k_{1}-k_{2}\right\vert
D\left( u,\widetilde{0}\right) .$
\end{theorem}
%%
\begin{definition}
\label{def4} (see \cite{Kaleva}). Let $f:[a,b]\rightarrow \mathbb{R}_{F}$:
%
\newline$(i)$ $f$ is fuzzy continuous at $x_{0}\in\lbrack a,b]$, if $\forall~\varepsilon>0$ there exists $\delta>0$
such that $\left\vert x-x_{0}\right\vert <\delta $ implies $D(f(x),f(x_{0}))<\varepsilon$.
%
\newline$(ii)$ $f$ is bounded iff there exists $M>0$ such that $D\left(  f\left(  x\right)  ,\widetilde {0}\right)  \leq M$,
 $\forall ~x\in\lbrack a,b].$
\end{definition}
%%
We note that $f$ is fuzzy continuous on $[a,b]$ if
$f$ is continuous at each $x_{0}\in\lbrack a,b]$, and the space of all
such functions is denoted by $C([a,b],\mathbb{R}_{F})$ and any fuzzy continuous function is bounded.
%
For any fuzzy-valued function (or fuzzy mapping) $f:[a,b]\rightarrow
\mathbb{R}_{F}$, the left and right $r-$level
functions of $f$ are specified respectively as $\underline{f}\left(  \cdot,r\right)
,$ $\overline{f}\left( \cdot,r\right) :[a,b]\rightarrow \mathbb{R}$, $r\in
\lbrack0,1]$.

The uniform distance between $f,~g\in C([a,b],\mathbb{R}_{F})$ is denoted by the supremum metric $D^{\ast}(f,g)=\sup\limits_{a\leq x\leq b}D(f(x),g(x))$
on the set $C([a,b],\mathbb{R}_{F})$.
%
The space $C([a,b],\mathbb{R}_{F})$ equipped with the uniform distance is a complete metric space.
%%
\begin{definition}
\label{def7} (see \cite{Gal}.) Let $f:[a,b]\rightarrow \mathbb{R}_{F}$ be a fuzzy-valued function.
For every  partition $P=\{[x_{i-1},x_{i}]:~i=\overline{1,n}\}$ of $[a,b$ and arbitrary point $\xi_{i}\in[x_{i-1},x_{i}],~i=\overline{1,n}$,
we say that $f$ is fuzzy-Riemann integrable to $I_f\in \mathbb{R}_{F}$ if
for $\forall$ $\varepsilon>0$, there exists $\delta>0$ such that $\max\{\left\vert x_{i}-x_{i-1}\right\vert :i=\overline{1,n}\}<\delta$
implies that
\[
D\left(
{\displaystyle\sum\limits_{i=1}^{n}}
\left(  x_{i}-x_{i-1}\right)  \odot f\left(  \xi_{i}\right)  ,I_f\right)
<\varepsilon.
\]
The quantity $I_f$ is written as%
\[
I_f=(FR)\int_{a}^{b}f(x)dx.
\]
\end{definition}
%
If $f\in C([a,b],\mathbb{R}_{F})$, then fuzzy-Riemann integral of $f$ exists, and
\[
\left[  (FR)\int_{a}^{b}f\left(  x\right)  dx\right] ^{r}=\left[\int_{a}%
^{b} \underline{f}\left( x,r\right) dx,\int_{a}%
^{b} \overline{f}\left( x,r\right) dx\right],\text{ \ }\forall ~0\leq r\leq 1.
\]
%
\begin{lemma} \label{Lem8}
(see \cite{Gal}). If $f,g\in C([a,b],\mathbb{R}_{F})$, then the
function $F:[a,b]\rightarrow \mathbb{R}_{+}$ given by $F(x)=D(f(x),g(x))$ is continuous
on $[a,b]$, and%
\[
D\left(  (FR)\int_{a}^{b}f(x)dx,(FR)\int_{a}^{b}g(x)dx\right)  \leq\int
_{a}^{b}D(f(x),g(x))dx.
\]
\end{lemma}
%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%
\begin{definition}
\label{def8} (see \cite{Bede}.) Let $f:[a,b]\rightarrow \mathbb{R}_{F}$
is $L$-Lipschitz. Then
\[D(f(x_1),f(x_2))\leq L|x_1-x_2|,\quad \forall~x_1,x_2 \in [a,b].\]
A function $F:\mathbb{R}_{F} \rightarrow \mathbb{R}_{F}$
is $L$-Lipschitz if there exists $L_F\geq 0$ such that $D(F(u_1),F(u_2))\leq L_F D(u_1,u_2)$
, $\forall$ $u_1,u_2 \in \mathbb{R}_{F}$.
\end{definition}
%According to \cite{Bede}, any Lipschitz function is continuous.
%%
\begin{definition}
\label{def8} (see \cite{Jiang}). Block-pulse functions on the unit interval $[a,b)$ is defined as follows:
\begin{equation*}\label{bp}
\varphi_i(t)=\left\{\begin{array}{l} {1} \qquad\qquad t\in [a+(i-1)h,a+ih), \\
{0 \qquad\qquad otherwise,} \end{array}\right.
\end{equation*}%
where $i=\overline{1,n}~$ with a positive integer value for $n$ and $h=\frac{b-a}{n}$. Also, $\varphi_i$ is called $i^{th}$ block-pulse function (BPF).
\end{definition}
Some properties of the BPFs are disjointness, orthogonality and completeness (see \cite{Jiang}).
\\
It is clear that $\varphi_i(t)\ge 0, $ for all $~t\in [a,b),~\varphi_1(t), \varphi_2(t) ,..., \varphi_n(t)$ are linearly independent, and
\begin{equation*}\label{unity}
\sum _{i=1}^n \varphi_i(t)=1.
\end{equation*}%
%%
Now, we employ the following notation:
\begin{equation}\label{intunity}
B_i(t)=\int_0 ^t \varphi_i(\tau)d\tau,\quad  t\in[a,b].
\end{equation}%
By using the definition of block-pulse function the integral given in Eq. \eqref{intunity}
can be computed as follows:
\begin{equation*}\label{intbp}
B_i(t)=\left\{\begin{array}{l} {0}    \qquad\qquad t\in [a,a+(i-1)h), \\
                               {t-ih} \qquad t\in [a+(i-1)h,a+ih), \\
                               {h     \qquad\qquad t\in [a+ih,b)} \end{array}\right.
\end{equation*}%
where $h=\frac{b-a}{n}.$
Now, we can evaluate the approximate value of the fuzzy integral of $f\in C([a,b],\mathbb{R}_{F})$
which satisfy in Lipschitz condition from $a$ to $t=t_i$, based on block-pulse functions as follows:
\begin{equation*}
(FR)\int_{a}^{t_i} f(s)ds\simeq I(f)=(FR)\int_{a}^{t_i}\sum_{j=1}^{n}f(z_j)\odot \varphi_j(s)ds,
\end{equation*}%
where $z_j=\frac{t_{j-1}+t_{j}}{2}$ for $j=\overline{1,n}$.
The error estimate of this approximation can be obtained as:
\begin{equation*}
\begin{split}
&D\left((FR)\int_{a}^{t_i} f(s)ds,I(f)\right)=\\
&=D\left((FR)\int_{a}^{t_i} \sum_{j=1}^{n}\varphi_j(s)\odot f(s)ds,(FR)\int_{a}^{t_i}\sum_{j=1}^{n}f(z_j)\odot \varphi_j(s)ds\right)\\
&\le \int_{a}^{t_i} \sum_{j=1}^{n}\varphi_j(s) D(f(s),f(z_j))ds\le \int_{a}^{t_i} \sum_{j=1}^{n}\varphi_j(s)L|s-z_j|ds.
\end{split}
\end{equation*}
By similar way in \cite{Ziari6}, we compute the above integral as follows:
%
\begin{equation*}
\begin{split}
&\int_{a}^{t_i}\sum _{j=1}^n\phi_{j}(s)\left(L\left|s-z_j\right|\right)ds=\\
&=L\sum _{j=1}^n\int_{a}^{t_i}\phi_{j}(s)\left|s-z_j\right|ds=L\sum _{j=1}^n\int_{t_{j-1}}^{t_j}\left|s-t_{j-1}\right|ds\\%
&=L\sum _{j=1}^n\left(\int_{t_{j-1}}^{z_j}\left(z_j-s\right)ds+\int_{z_j}^{t_j}\left(s-z_j\right)ds\right)%
=\frac{L(b-a)^2}{4n}.
\end{split}
\end{equation*}%
Finally, we have:
%
\begin{equation*}\label{errf}
D\left((FR)\int_{a}^{t_i} f(s)ds,(FR)\int_{a}^{t_i}\sum_{j=1}^{n}f(z_j)\odot \varphi_j(s)ds\right) \leq \frac{L(b-a)^2}{4n}.
\end{equation*}
%%
\section{Successive approximations method}

In this section, problem \eqref{eq1} will be studied according to the following hypotheses:

(\textit{i}) $g\in C([a,b],\mathbb{R}_{F})$ and $f\in C\left([a,b]\times [a,b] \times \mathbb{R}_{F},\mathbb{R}_{F}\right)$;

(\textit{ii}) there exist $\alpha$ and $\eta>0$ such that%
\[
D\left(  f\left(  t,s_1,u_1\right)  ,f\left(  t, s_2 ,u_2\right)  \right)  \leq\eta |s_1-s_2|
+\alpha D\left(  u_1,u_2\right) ,~\forall \text{ }t,s_1,s_2\in\lbrack
a,b],~u_1,u_2\in \mathbb{R}_{F};
\]

(\textit{iii}) $\gamma=\alpha(b-a)<1;$

(\textit{iv}) there exists $\beta> 0$ such that%
\[D(g(t_1),g(t_2))\leq \beta |t_1-t_2|,~\forall ~t_1,t_2\in[a,b].\]
%
\begin{lemma}\label{conti-lem}
Suppose that the following assumptions are satisfied:\\

(\textit{i}) $f\in C\left([a,b]\times [a,b] \times \mathbb{R}_{F},\mathbb{R}_{F}\right)$,

(\textit{ii}) there exists $\mu>0$ such that%
\[
D\left(  f\left(  t_1,s,u\right)  ,f\left(  t_2, s ,u\right)  \right)  \leq\mu |t_1-t_2|
,~\forall \text{ }t_1,t_2,s\in\lbrack
a,b],~u\in \mathbb{R}_{F},
\] then the function $F:[a,b]\rightarrow \mathbb{R}_{F}$ formulated by
\[F(t)=(FR)\int_a^t f(t,s,y(s))ds,\] is uniformly continuous.
\end{lemma}
%

\begin{proof}
By the hypothesis $(i)$, there exist constants $M_y,~M_f>0$ such that
$D\left(y(s),\tilde{0}\right)\le M_y,~D\left(f(t,s,u),\tilde{0}�\right)\le M_f$ for all $t,s\in\lbrack
a,b],~u\in \mathbb{R}_{F}$.\\
Taking $t_1,t_2\in \lbrack a,b]$ with $a\leq t_{1}\leq t_{2}\leq b$, we have: and regarding to hypothesis $(ii)$  we have:
%
\begin{equation*}
\begin{split}
D(F(t_1),F(t_2))& =D\left( (FR)\int_{a}^{t_1}f(t_1,s,y(s))ds,(FR)\int_{a}^{t_2}f(t_2,s,y(s))ds\right) \\
& \leq D\left( (FR)\int_{a}^{t_1}f(t_1,s,y(s))ds,(FR)\int_{a}^{t_1}f(t_2,s,y(s))ds\right)\\
&\quad +D\left( (FR)\int_{t_1}^{t_2}f(t_2,s,y(s))ds,\tilde {0}\right) \\
& \leq \int_{a}^{t_1}D\left( f(t_1,s,y(s)),f(t_2,s,y(s))\right) ds\\
&\quad +\int_{t_1}^{t_2}D\left(f(t_2,s,y(s)),\tilde{0}\right)ds.\\
\end{split}%
\end{equation*}%
According to hypothesis $(ii)$, we obtain:
%
\begin{equation*}
D(F(t_1),F(t_2)) \leq \int_{a}^{t_1}\mu |t_1-t_2|ds+\int_{t_1}^{t_2}M_fds\leq \mu(b-a)|t_1-t_2|+M_f(t_2-t_1).
\end{equation*}%
%
For any $\epsilon >0$, there exists $\delta$,
\begin{equation*}
0<\delta\leq \min\left(1,\frac{\epsilon }{\mu(b-a)+M_f}\right)
\end{equation*}%
such that $|t_1-t_2|<\delta$ concludes that
\begin{equation*}
D(F(t_1),F(t_2)) \leq \left(\mu(b-a)+M_f\right)\delta\leq \epsilon.
\end{equation*}%
%
This proves that $F$ is uniformly continuous.
\end{proof}

%%
To express the existence and uniqueness of Eq. \eqref{eq1},
we consider the associated sequence of successive approximations:
\begin{equation*} \label{F-itr}
\begin{split}
&y_{0}(t)=g(t),\\
&y_{m}(t)=g(t)\oplus(FR)\int_{a}^{t} f(t,s,y_{m-1}(s))ds,\quad
t\in\lbrack a,b],\text{ }m\in\mathbb{N} %
\end{split}
\end{equation*}%
and we construct the sequence $(F_m(t))_{m\in\mathbb{N}}$ such that
$F_m:[a,b]\times [a,b]\rightarrow \mathbb{R}_{F},\text{ by } F_m(t,s)= f(t,s,y_m(s)),~ m\in\mathbb{N}.$
%%
\begin{theorem}(see \cite{Bica3}.) Considering hypotheses $ (i)-(iii)$, the equation (1) has unique solution in $ C([a,b],\mathbb{R}_{F}),y^* \in C([a,b],\mathbb{R}_{F})$
and sequence of Picard's approximations $(y_{m})_{m \in \mathbb{N}} \subset C([a,b],\mathbb{R}_{F}) $, depicted in (3) tends to $ y^*$ in $ C([a,b],\mathbb{R}_{F})$ for any election of $ y_{0} \in C([a,b],\mathbb{R}_{F})$. Furthermore, the below error bounds satisfied:
\begin{equation}\label{bnd1}
    D(y^*(t),y_{m}(t)) \le \frac{\gamma ^m}{1- \gamma}.D(y_{1}(t),y_{0}(t),\quad
    \forall t \in [a,b],m \in \mathbb{N^*},
\end{equation}
and
\begin{equation*}
     D(y^*(t),y_{m}(t)) \le \frac{\gamma }{1- \gamma}.D(y_{m}(t),y_{m-1}(t),\quad
    \forall t \in [a,b],m \in \mathbb{N^*}.
\end{equation*}
    Choosing $ y_{0}\in C([a,b],\mathbb{R}_F)$, $y_{0}=g$ the inequality \eqref{bnd1} converts
\begin{equation}\label{errest}
     D(y^*(t),y_{m}(t)) \le \frac{\gamma ^m}{1- \gamma}M_{0}(b-a),\quad
     \forall t \in [a,b],m \in \mathbb{N^*},
\end{equation}
    where $M_{0} \ge 0 $ is such that $ D(f(t,s,g(s)),\tilde 0) \le M_{0}$, $  \forall ~ t,s \in [a,b] $, Furthermore, conditions (i)-(iv) provide  uniformly bounded and uniform Lipschitz of the sequences $(y_{m})_{m \in \mathbb{N^*}} $ and $(F_{m})_{m \in \mathbb{N^*}}$ respectively.
\end{theorem}
%%
\section{Description of the method and error analysis}

Now, we present the iterative method using block-pulse functions to approximate the solution of \eqref{eq1}.
Taking into account the uniform partition of the interval $[a,b]$,%
\[
\Delta:a=t_{0}<t_{1}<...<t_{n-1}<t_{n}=b,
\]
with  $t_{i}=a+\frac{i(b-a)}{n},\quad\forall~ i=\overline{0,n}$.
For $i=\overline{0,n}$, we have:
\begin{equation}\label{itr1}
\begin{split}
&y_{0}(t_i)=g(t_i),\\
&y_{m}(t_i)=g(t_i)\oplus(FR)\int_{a}^{t_i} f(t,s,y_{m-1}(s))ds,\quad
m\in\mathbb{N}. %
\end{split}
\end{equation}%
Now, we approximate the integral in \eqref{itr1} as follows:
\begin{equation*}
(FR)\int_{a}^{t_i} f(t,s,y_{m-1}(s))ds\simeq (FR)\int_{a}^{t_i}\sum_{j=1}^{n}f(t_i,z_j,y_{m-1}(z_j))\odot \varphi_j(s)ds,
\end{equation*}%
where $z_j=\frac{t_{j-1}+t_{j}}{2}$ for $j=\overline{1,n}$.\\
%
Then, the following approximations are gained:
\begin{equation}\label{itr2}
y_{m}(t_i)=g(t_i)\oplus \sum_{j=1}^{n} f(t_i,z_j,y_{m-1}(z_j))\odot B_j(t_i)\oplus R_{m,i},
\end{equation}%
where
\begin{equation}\label{itr3}
B_j(t_i)=\int_{a}^{t_i}\varphi_j(s)ds\quad \text{and} \quad D(R_{m,i},\tilde 0)\le \frac{L(b-a)^2}{4n}.
\end{equation}%
In this way, we get the following iterative procedure:
\begin{equation}\label{itr4}
\begin{split}
&\tilde{y}_{0} (t_i)=y_{0}(t_i)=g(t_i),\\
&\tilde{y}_{m}(t_i)=g(t_i)\oplus \sum_{j=1}^{n} f(t_i,z_j,\tilde{y}_{m-1}(z_j))\odot B_j(t_i).
\end{split}
\end{equation}%
%%
The following theorem proves the convergence of the presented method.
\begin{theorem}
By the hypothesis (i)-(iv), the iterative scheme (\ref{itr4}) converges
to the exact solution $y^{\ast}$ of (\ref{eq1}), and the following error bound is true on the knots
$t_i=\frac{i(b-a)}{n},\text{ } i=\overline{0,n}$:
\begin{equation*} \label{eq9}%
D(y^{\ast}(t_{i}),\tilde{y}_{m}(t_{i}))\leq \frac{\gamma ^m}{1- \gamma}M_{0}(b-a)+\frac{L(b-a)^2}{4n(1-\gamma)}.
\end{equation*}
\end{theorem}
%
\begin{proof} % Proof of Theorem
According to Eq. \eqref{errest}, we know:
\begin{equation}\label{err1}
     D(y^*(t_i),y_{m}(t_i)) \le \frac{\gamma ^m}{1- \gamma}M_{0}(b-a),\quad
     \forall i=\overline{0,n},~m \in \mathbb{N^*}.
\end{equation}
Since
\begin{equation}\label{ineq1}
D(y^{\ast}(t_{i}),\tilde{y}_{m}(t_{i}))\leq D(y^{\ast
}(t_{i}),y_{m}(t_{i}))+D(y_{m}(t_{i}),\tilde{y}_{m}(t_{i})),
\end{equation}
so, we try to gain the estimates for
$D(y_{m}(t_{i}),\tilde{y}_{m}(t_{i}))$ , $\forall ~i=\overline{0,n},~m \in \mathbb{N^*}$.
\\
Since%
\begin{equation*}
y_{1}(t_{i})=g(t_{i})\oplus\sum_{j=1}^{n}f(t_i,z_j,g(z_j))\oplus R_{1,i},\text{ }i=\overline{0,n}, \label{fmi}%
\end{equation*}
Hence
%
\begin{equation*}
\begin{split}
&D(y_{1}(t_{i}),\tilde{y}_{1}(t_{i})) \leq  D(g(t_{i}),g(t_{i}))\\
&+D\left(\sum_{j=1}^{n}f(t_i,z_j,g(z_j))\odot B_j(t_i)\oplus R_{1,i},\sum_{j=1}^{n} f(t_i,z_j,g(z_j))\odot B_j(t_i)\right)\\
&\le  D\left(R_{1,i},\tilde 0\right)\le\frac{L(b-a)^2}{4n}.
\end{split}
\end{equation*}
%
Now, using Eqs. \eqref{itr2}-\eqref{itr4} we get:
\begin{equation*}
\begin{split}
&D(y_{2}(t_{i}),\tilde{x}_{2}(t_{i})) \leq  D(g(t_{i}),g(t_{i}))+\\
&+D\left(\sum_{j=1}^{n}f(t_i,z_j,y_{1}(t_{j}))\odot B_j(t_i)\oplus R_{2,i},\sum_{j=1}^{n} f(t_i,z_j,\tilde{y}_{1}(t_{j}))\odot B_j(t_i)\right)\\
&=  D\left(\sum_{j=1}^{n}f(t_i,z_j,y_{1}(t_{j}))\odot \int_{a}^{t_i}\varphi_j(s)ds\oplus R_{2,i},\sum_{j=1}^{n} f(t_i,z_j,\tilde{y}_1(z_j))\odot \int_{a}^{t_i}\varphi_j(s)ds\right)\\
&\le \int_{a}^{t_{i}}\sum_{j=1}^{n}\varphi_j(s) D\left( f(t_i,z_j,y_1(z_j)), f(t_i,z_j,\tilde{y}_1(z_j))\right)ds+D\left(R_{2,i},\tilde 0\right)\\
&\le \int_{a}^{t_{i}}\sum_{j=1}^{n}\varphi_j(s)\alpha D\left(  y_1(z_j),\tilde{y}_1(z_j)\right)ds+D\left(R_{2,i},\tilde 0\right)\\
&\le \frac{\alpha L(b-a)^3}{4n}+\frac{L(b-a)^2}{4n}=\left(1+\gamma \right)\frac{L(b-a)^2}{4n}.\\
\end{split}
\end{equation*}
By mathematical induction, for $m\in \mathbb{N}^\ast,\text{ } m\ge 3$, we get:
\begin{equation*}
\begin{split}
D(y_{m}(t_{i}),\tilde{y}_{m}(t_{i})) &\leq \left( 1+\gamma\cdots+\gamma^{m-1}\right)\frac{L(b-a)^2}{4n}\\
&= \left(\frac{1-\gamma^m}{1-\gamma} \right)\frac{L(b-a)^2}{4n}.
\end{split}
\end{equation*}
By considering condition (iii), we get:
\begin{equation}\label{err3}
\begin{split}
D(y_{m}(t_{i}),\tilde{y}_{m}(t_{i})) \leq \frac{L(b-a)^2}{4n(1-\gamma)},\quad\forall ~i=\overline{0,n},~m \in \mathbb{N^*}.
\end{split}
\end{equation}
So, the assertion follows from \eqref{err1}, \eqref{ineq1} and \eqref{err3}.
\end{proof}
%% End of Proof
\section{Numerical experiment}

To establish the theoretical results in the previous section, a numerical example
has been demonstrated by iterative procedure \eqref{itr4}.
The results have been provided by Mathematica.

%
\begin{example}
\label{exmp1}  Let the NFVIE
\begin{equation*}
y(t)=g(t)\oplus(FR)\int_{0}^{t} t e^{-s}\odot {[y(s)]^2}ds,\quad
s\le t, \quad t\in\lbrack0,{1}]
\end{equation*}
with
\begin{equation*}
\underline{g}\left(  t,r\right) =r{t}-tr^2+r^2 e^{-t}\left(\frac{t^3}{2}+t^2+t\right),
\end{equation*}%
\begin{equation*}
\overline{g}\left(  t,r\right) =(2-r){t}+(2-r)^2 \left(\frac{t^3e^{-t}}{2}+t^2e^{-t}+te^{-t}-t\right).
\end{equation*}
The exact solution is as follows%
\begin{equation*}
\left[\underline{y}(t,r),\overline{y}(t,r)\right]=\left[r {t},(2-r) {t}\right].
\end{equation*}
To solve the above stated equation, we apply the iterative scheme \eqref{itr4} with $n=10$ and $m=5$,
to determine $r$-levels of the results
$\underline{e}({r})=|\underline{\tilde{y}_m}(t,r)-\underline{y}(t,r)|,$ $\overline{e}({r})=|\overline{\tilde{y}_m}(t,r)-\overline{y}(t,r)|,$
at point $t=0.5$, for $r\in\{\frac{1}{4}%
,\frac{1}{2},\frac{3}{4},1\}$ are put in Table 1.
To peruse the convergence of the presented method is considered $n=100$ and $m=5$ too and the obtained results are given in Table 1.

%%
\begin{table}[!ht]
\centering
\caption{Error on the $r$-level sets for Example \ref{exmp1} at $t=0.5$}\label{vtab1}
  \begin{tabular}{c c c c c }
   \hline
   &{ \qquad $n=10,~~m=5~$ \qquad} & { \qquad $n=100,~~m=5~$ \qquad }& \\
   \hline
    $r$-level &  $\qquad \underline{e}({r}) \qquad$   $\quad \overline{e}({r}) \qquad$  &
    $\qquad \underline{e}({r})\qquad $  $ \overline{e}({r}) $\\
   \hline
    %
    0.25 & 1.252e-004 ~ 6.392e-003 & 1.196e-005 ~ 6.031e-004\\
    %
    0.50 & 5.042e-004 ~ 4.664e-003 & 4.808e-005 ~ 4.410e-004\\
    %
    0.75 & 1.142e-003 ~ 3.216e-003 & 1.087e-004 ~ 3.048e-004\\
    %
    1    & 2.045e-003 ~ 2.045e-003 & 1.941e-004 ~ 1.941e-004\\
   \hline
  \end{tabular}\\
 \end{table}
%%
According to Table 1, the convergence of presented method is confirmed.
%%
\end{example}
%---------------------------------------
\section{Conclusions}

The current paper presented an iterative numerical scheme via block-pulse functions
to solve NFVIEs. The proposed method is computationally efficient with respect to the other existing methods.
Moreover, We have given the error estimate of the iterative technique by taking into account the Lipschitz
condition. The numerical results confirmed the validity of the results in theoretical aspect.
%%




{\small

\noindent{\bf Kamran Akhavan Zakeri}

\noindent PhD student of Applied Mathematics

\noindent Department of Mathematics

\noindent Central Tehran Branch,

\noindent Islamic Azad University

\noindent Tehran, Iran

\noindent E-mail: kamran.71722002@yahoo.com}\\

{\small
\noindent{\bf  Mohammad Ali Fariborzi Araghi  }

\noindent Associate Professor of Applied Mathematics

\noindent  Department of Mathematics

\noindent Central Tehran Branch,

\noindent Islamic Azad University

\noindent Tehran, Iran

\noindent E-mail: fariborzi.araghi@gmail.com}\\

{\small
\noindent{\bf  Shokrollah Ziari }

\noindent Assistant Professor of Applied Mathematics

\noindent Department of Mathematics,

\noindent Firoozkooh Branch,

\noindent Islamic Azad University,

\noindent Firoozkooh, Iran

\noindent E-mail: shok\_ziari@yahoo.com}\\

\end{document}

